Author's response to reviews

Title: Benign prostatic hyperplasia increases bladder cancer risk in diabetic patients

Authors:

Chin-Hsiao Tseng (ccktsh@ms6.hinet.net)

Version: 2 Date: 24 August 2012

Author's response to reviews:

August 24, 2012

Dear Editor,

Uploaded please find a manuscript entitled: “Benign prostatic hyperplasia significantly increases bladder cancer risk in diabetic patients” for your kind consideration of possible publication.

While submitting the manuscript, I would like to declare that:

1) The author of this research paper has directly participated in the planning, execution, or analysis of the study.
2) The author of this paper has read and approved the final version submitted.
3) The contents of this manuscript have not been copyrighted or published previously.
4) The contents of this manuscript are not now under consideration for publication elsewhere.
5) The contents of this manuscript will not be copyrighted, submitted, or published elsewhere while acceptance by the Journal is under consideration.
6) There are no directly related manuscripts or abstracts, published or unpublished, by the author of this paper.

Sincerely yours,

Chin-Hsiao Tseng, MD, PhD